**Author's response to reviews**

**Title:** A pilot study of Aboriginal health promotion from an ecological perspective

**Authors:**
- Rachel E Reilly (rachel@unimelb.edu.au)
- Marion Cincotta (marionc@vaccho.com.au)
- Joyce Doyle (jdoyle6@bigpond.com)
- Bradley R Firebrace (bradleyf@unimelb.edu.au)
- Margaret Cargo (margaret.cargo@unisa.edu.au)
- Gemma Van den Tol (gvandentol@gvct.com.au)
- Denise Morgan-Bulled (denise@vineymorganams.com.au)
- Kevin G Rowley (rowleyk@unimelb.edu.au)

**Version:** 4  **Date:** 8 August 2011

**Author's response to reviews:** see over
8th August, 2011

Dear Editors,

RE: Reilly et al. A pilot study of Aboriginal health promotion from an ecological perspective.

Thank you for accepting this paper for publication (in principle). As requested, we have made the following amendments:

Major revisions:
- Author list: The author list, including initials, is now written in the manuscript as they appear in the submission system.
- Figure title: The figure title appears towards the bottom of page 8.
- The keywords appearing at the end of the abstract have been removed (Keywords: Health promotion; ecological theory; evaluation; Aboriginal and Torres Strait Islander health).

Minor revisions:
- Tracking has been removed from the document.
- A cropped and re-formatted figure has been uploaded.
- Every effort has been made to correct minor typographical errors.

Kind regards,

Rachel Reilly